Author’s response to reviews

Title: Sudden infant death syndrome and prenatal maternal smoking: Rising attributed risk in the Back to Sleep era

Authors:

Mark E Anderson (manderso@dhha.org)
Daniel C Johnson (daniel.johnson@uchsc.edu)
Holly A Batal (hbatal@dhha.org)

Version: 6 Date: 3 January 2005

Author’s response to reviews: see over
RE: MS 1627615283380851
“Sudden Infant Death Syndrome and Prenatal Maternal Smoking: Rising Attributed Risk in the Back to Sleep Era”
January 3, 2005

Dear Editorial Staff:

I have made the following formatting changes to the manuscript as requested:

I have added a sentence to the abstract background
I removed the titles of the tables from the spreadsheet portion of the tables
Figure 1 has been modified to reduce blank space

Please contact me if additional changes are needed.

Mark E. Anderson MD FAAP
Assistant Professor of Pediatrics
School of Medicine, University of Colorado
Community Health Services, Division of Pediatrics
Denver Health
777 Bannock Street, Mail Code 1911
Denver, Colorado  80204
303 436 4098
manderso@dhha.org